\begin{table}%T2
%\centering
\par
\caption{Mass determination for the companion ($M_{\rm j}$ for Jupiter mass).}\label{t2}
\begin{tabular}{lll}
\hline\hline
Model                  & Parameters used        & Mass \\ \hline
Burrows et al. (\cite{Burrows97}), Fig.~7 & 35~Myr, $L$ at 85 pc    & 20--30 $M_{\rm j}$ \\
Burrows et al. (\cite{Burrows97}), Fig. 9 & $T=2300$~K, 35~Myr    & 15--35 $M_{\rm j}$ \\
Burrows et al. (\cite{Burrows97}), Fig. 11& $T=2300$~K, $L$ at 85 pc & 20--40 $M_{\rm j}$ \\
Chabrier et al. (\cite{Chabrier00}), Fig. 2 & $T=2300$~K, 35~Myr    & 20--40 $M_{\rm j}$ \\
Chabrier et al. (\cite{Chabrier00}), Table 1 & $T$, $L$ and $M_{K}$ at 85 pc & 35--45 $M_{\rm j}$ \\
Baraffe et al. (\cite{Baraffe02}), Fig. 2 & $T$, 35~Myr, $L$ at 85 pc & ${\sim} 20$~$M_{\rm j}$ \\
Baraffe/Chabrier in C03     & $M_{K}$ vs. $J-K$ CMD    & 20--40 $M_{\rm j}$ \\
Wuchterl \& Tscharnuter (\cite{Wuchterl03})    & $T$, $L$ at 85 pc, 35~Myr & ${\sim} 7$~$M_{\rm j}$ \\
D'Antona \& Mazzitelli (\cite{DAntona98})     & $T$, $L$ at 85 pc (Fig. 5) & 20--50 $M_{\rm j}$ \\ \hline
\end{tabular}
%\vspace{4mm}
\end{table}